Author’s response to reviews

Title: SCL20A2 mutation presenting with acute ischemic stroke: a case report

Authors:

xiaooyu zhang (cyzxy731@126.com)
Gaoting Ma (demo_doctor@163.com)
Zhangning Zhao (zzn304@aliyun.com)
Meijia Zhu (zmjqfsyy@126.com)

Version: 2 Date: 02 Jan 2018

Author’s response to reviews:

Thank you for your useful comments and suggestions on our manuscript (NURL-D-17-00460R1). We have modified the manuscript accordingly, and detailed corrections are listed below point by point:

1) Please submit a clean version of the manuscript so that track changes is turned off.

Reply: track changes have been turned off in the manuscript.

2) Please use authors’ initials for Authors’ contributions.

Reply (line 212-214): XY Z and GT M examined the patient and wrote the manuscript. XY Z and ZN Z analyzed neuroimagings. MJ Z participated in the design of the case report and helped to draft the manuscript. All authors read and approved the final manuscript.